Author's response to reviews

Title: The Nigerian national blindness and visual impairment survey: rationale, objectives and detailed methodology

Authors:

   Dineen B (brendan.dineen@nuigalway.ie)
   Gilbert CE (Clare.Gilbert@lshtm.ac.uk)
   Rabiu MM (mrabiu@hotmail.com)
   Kyari F (fatimaygk@yahoo.co.uk)
   Mahdi MM (drtafida@yahoo.com)
   Abubakar T (abdullmm@yahoo.com)
   Christian E (ccezelum@yahoo.com)
   Gabriel E (entekume_66@yahoo.com)
   Elhassan E (elizabethelhassan@yahoo.co.uk)
   Abiose A (adenikeabioseo@yahoo.com)
   Faal H (hfaal@sightsavers.org)
   Jiya J (jiyajy@yahoo.com)
   Ozemela C (cpozems@yahoo.co.uk)
   Lee PS (pak.lee@ucl.ac.uk)
   Murthy V S Gudlavalleti (gvsmurthy2000@yahoo.com)

Version: 6 Date: 10 September 2008

Author's response to reviews:

The following revisions have been made in the manuscript:

1. Sentence case has been used in the title except for the first word and proper noun (Nigeria).
2. All references have been placed before the full stop in square brackets.
3. Reference numbers 30 and 31 have been repositioned.
4. There are no blank pages in the document.

regards,
Gudlavalletti Murthy